CSci2111: Assignment and Project Set # 1 Handed Out on: Thursday January 20, 2000 Due Date: Thursday February 3, 2000 1 Written Exercises

ffl Problem 6 on p.41 (Chapter 2) ffl Problem 11 on p. 112 (Chapter 3)

2 Programming Exercises

Use Java or C++ to do the following:

ffl Problem 10 on p. 41 (Chapter 2) ffl Problem 12 on p. 42 (Chapter 2)

3 Programming Project, Part I (of V)

Use Java or C++ to do the following (if using Java, discard any comment (from the problem question) regarding the use of C++):

ffl Problems 1, 2, and 3 on p. 12 (Chapter 1) ffl Problems 14 and 15 on p. 42 (Chapter 2)